Author's response to reviews

Title: Impact of movement training on upper limb motor strategies in persons with shoulder impingement syndrome

Authors:

Jean-Sébastien Roy (jean-sebastien.roy.1@ulaval.ca)
Hélène Moffet (helene.moffet@rea.ulaval.ca)
Bradford J McFadyen (brad.mcfadyen@rea.ulaval.ca)
Richard Lirette (rlirette@mediom.qc.ca)

Version: 3 Date: 12 May 2009

Author's response to reviews:

Three small changes were brought to the manuscript:

Background, 1st paragraph (1st sentence): a space was added between "deficits" and the references.

Background, last paragraph (3rd line): "using muscular activation..." was changed for "using such variables (muscular activation...)

Abbreviations, E4: a space was added between "evaluation" and "24"